Author’s response to reviews

Title: Applying evidence-based medicine in general practice: a video-stimulated interview study on workplace-based observation
Authors:
Lisanne Welink (l.s.welink@umcutrecht.nl)
Kaat Van Roy (kaatje.vanroy@ugent.be)
Roger Damoiseaux (r.a.m.j.damoiseaux@umcutrecht.nl)
Hilde Suijker (hasuijker@gmail.com)
Peter Pype (Peter.Pype@ugent.be)
Esther de Groot (E.deGroot@umcutrecht.nl)
Marie-Louise Bartelink (M.E.L.Bartelink@umcutrecht.nl)

Version: 3 Date: 20 Dec 2019

Author’s response to reviews:

Dear Megan Barling / Gerard Clunn,
Please find attached our revised manuscript entitled "Applying evidence-based medicine in general practice: a video-stimulated interview study on workplace-based observation" (FAMP-D-19-00315). On behalf of all the authors, I would like to thank you again for giving us the opportunity to revise our manuscript for publication in BMC Family Practice. Below we explain how we have addressed the editor’s comments and requests. We hope that the revised manuscript will be acceptable for publication and are looking forward to hearing from you.
Yours sincerely,
Also on behalf of the co-authors,
Lisanne S. Welink, MD
Julius Center for Health Sciences and Primary Care, University Medical Center Utrecht
Universiteitsweg 100
3584 CX Utrecht
Unluckily BMC does not accept boxes for publication. Please remove Box 1 and include the information in a different format. This comment about Box 1 was rather unclear for us, because our manuscript did not include a ‘box’. We asked for further explanation by e-mail, but did not receive a response so far. Due to the tight deadline of revision, we for now guessed that this comment concerned Table 1, which we transformed into Figure 1. Please let us know if we misinterpreted the comment.

2. Consent

Thank you for specifying the physicians provided written consent. Please update this under 'ethics approval and consent to participate'. We updated this under ‘ethics approval and consent to participate’ (page 18, line 440) --Please specify whether the ethics committee approved the verbal consent obtained from the patients and also state why this method of consent was chosen over written consent under 'ethics approval and consent to participate'. We specified this under ‘ethics approval and consent to participate’. (page 18, line 442-447)

3. Cite

Please ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication. To our knowledge, all figures/tables/supplementary files are cited within the text. We adjusted the numbers of tables and figures and the citations within the text, based on the adjustments we made following comment 1.

4. Clean Manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text. We uploaded a clean, black-and-white version of Table 1 and 2. Furthermore, we uploaded a new, clean version of the manuscript.